Author's response to reviews

Title: Tracking of eating patterns and overweight - a follow-up study of Norwegian school children from middle childhood to early adolescence

Authors:

Inger M Oellingrath (inger.m.oellingrath@hit.no)
Martin V Svendsen (martin.svendsen@sthf.no)
Anne Lise Brantsaeter (anne.lise.brantsaeter@fhi.no)

Version: 5 Date: 5 October 2011

Author's response to reviews:

In this version we have edited the manuscript as follows:

Page 11, Results: "low factor loadings (<0.30)" changed to "low factor loadings (between -0.30 and +0.30)"

Page 14, Results. In the second paragraphs (Table 4) corrected to (Table 3).

Page 17, "in different populations changed to "in a nationwide study of Norwegian adolescents"

Table 1, below the table: factor "loadings above -/+0.25" changed to "factor loadings >0.25 or <-0.25"

All tables: Removed excessive horizontal lines.

References: Nordic letters in names in References 3, 6, 18, 40, 41 and 44 have been edited.

Thank you.